Author's response to reviews

Title: Disrupting the rhythm of depression: design and protocol of a randomized controlled trial on preventing relapse using brief cognitive therapy with or without antidepressants

Authors:

Claudi LH Bockting (c.l.h.bockting@rug.nl)
Hermien J Elgersma (h.j.elgersma@rug.nl)
Gerard D van Rijsbergen (g.d.van.rijsbergen@rug.nl)
Peter de Jonge (peter.de.jonge@med.umcg.nl)
Hans Ormel (j.ormel@med.umcg.nl)
Erik Buskens (e.buskens@epi.umcg.nl)
A. Dennis Stant (a.d.stant@epi.umcg.nl)
Peter J de Jong (p.j.de.jong@rug.nl)
Frenk PML Peeters (f.peeters@maastrichtuniversity.nl)
Marcus JH Huibers (m.huibers@maastrichtuniversity.nl)
Arnoud Arntz (arnoud.arntz@maastrichtuniversity.nl)
Peter Muris (muris@fsw.eur.nl)
Willem A Nolen (w.a.nolen@psy.umcg.nl)
Aart H Schene (a.h.schene@amc.uva.nl)
Steven D Hollon (steven.d.hollon@vanderbilt.edu)

Version: 3 Date: 7 January 2011

Author's response to reviews:

7th January 2011

Dear editor,

Herewith I would like to submit the revision of our paper entitled 'Disrupting the rhythm of depression: design of a randomized controlled trial on preventing relapse using brief cognitive therapy with or without antidepressants'.

Revision includes changes in the author list making it in line with the submission system. A contact author has been provided. All et al’s from the reference list were removed and the text references have now been cited by square brackets. A submitted reference has been removed from the reference list and is now mentioned in the text. Furthermore the table has been removed from the submission system and is now following the references. The title 'Figure' has been removed from the flow-chart figure and its location in the text has been removed having it now only being referred to in the text. The additional file (ethical approval) has been removed in the submission system.

We thank you for your comments and for considering our manuscript.

Yours truly,
Prof. dr. CLH Bockting